Article type:

Submitted version - Preprint

Link for final version:

https://www.thelancet.com/journals/laneur/article/PIIS1474-4422(16)30035-7/fulltext

This is a submitted-preprint version of an accepted manuscript. Note that revisions and technical editing may introduce changes to the manuscript text and/or graphics which could affect content. To access to the final version, click the link above.

Stakeholder cooperation to overcome challenges in orphan medicine development: the example of Duchenne muscular dystrophy

Volker Straub, Pavel Balabanov, Kate Bushby, Monica Ensini, Nathalie Goemans, Annamaria De Luca, Alejandra Pereda, Robert Hemmings, Giles Campion, Edward Kaye, Virginia Arechavala-Gomeza, Aurelie Goyenvalle, Erik Niks, Olav Veldhuizen, Pat Furlong, Violeta Stoyanova-Beninska, Matthew J Wood, Alex Johnson, Eugenio Mercuri, Francesco Muntoni, Bruno Sepodes, Manuel Haas, Elizabeth Vroom, Annemieke Aartsma-Rus

John Walton Muscular Dystrophy Research Centre and MRC Centre for Neuromuscular Diseases, Institute of Genetic Medicine, Newcastle University, Newcastle upon Tyne, UK(V Straub MD, K Bushby MD, MEnsini PhD, O Veldhuizen, A Aartsma-Rus PhD); European Medicines Agency, Human Medicines Evaluation Division, Scientific and Regulatory Management Department, CNS and Ophthalmology Office, London, UK (P Balabanov PhD), M Haas PharmD); Department of Child Neurology, University Hospitals Leuven, Leuven, Belgium (N Goemans MD); Unitof Pharmacology, Department of Pharmacy—Drug Sciences, University of Bari, Bari, Italy(A De Luca PhD); Duchenne Parent Project, Madrid, Spain (A Pereda); Medicines and Healthcare Product Regulatory Agency, London, UK (R Hemmings); BioMarin Pharmaceutical, DevelopmentalSciences, Novato, CA, USA (G Campion MD); SareptaTherapeutics, Cambridge, MA, USA (EKaye MD); Neuromuscular DisordersGroup, BioCruces HealthResearch Institute, Barakaldo, Spain(V Arechavala-Gomeza PhD); Université de Versailles SaintQuentin en Yvelines, Inserm U1179, Montigny le Bretonneux, France (A Goyenvalle PhD); LeidenUniversity Medical Center, Leiden, Netherlands (E Niks MD, A Aartsma-Rus); Parent Project Muscular Dystrophy, Hackensack, NJ, USA (P Furlong); Medicines Evaluation Board, Utrecht, Netherlands(V Stoyanova-Beninska MD); Department of Physiology, Anatomy and Genetics, University of Oxford, Oxford, UK (M JWood PhD); Joining Jack, Wigan, UK (A Johnson); Pediatric Neurology, Catholic University, Rome, Italy (E Mercuri MD); Dubowitz Neuromuscular Centre and MRC Centre for Neuromuscular Disease, University College London Institute of Child Health and Great Ormond Street Hospital, London, UK (F Muntoni MD); Faculdade de Farmácia, Universidade de Lisboa, Lisbon, Portugal (B Sepodes PhD); and United Parent Projects MuscularDystrophy, Amsterdam, Netherlands (E Vroom DMD)

Correspondence to:

Dr Annemieke Aartsma-Rus, Department of Human Genetics, Leiden University Medical Center, 2333 ZA Leiden, Netherlands a.m.rus@lumc.nl

Duchenne muscular dystrophy is a rare, progressive, muscle-wasting disease leading to severe disability and premature death. Treatment is currently symptomatic, but several experimental therapies are in development. Implemented care standards, validated outcome measures correlating with clinical benefit, and comprehensive information about the natural history of the disease are essential for regulatory approval of any treatment. However, for Duchenne muscular dystrophy and other rare diseases, these requirements are not always in place when potential therapies enter the clinical trial phase. A cooperative eff ort of stakeholders in Duchenne muscular dystrophy-including representatives from patients' groups, academia, industry, and regulatory agencies—is aimed at addressing this shortfall by identifying strategies to overcome challenges, developing the tools needed, and collecting relevant data. An open and constructive dialogue among European stakeholders has positively aff ected development of treatments for Duchenne muscular dystrophy; this approach could serve as a paradigm for development of treatments for rare diseases in general.

Introduction

Developing treatments for genetic diseases poses unique challenges, as shown by the example of Duchenne muscular dystrophy,

a rare, progressive, muscle-wasting disease affecting about one in 5000 newborn boys.¹²Duchenne muscular dystrophy is caused by mutations abolishing production of the muscle fibre-stabilising protein dystrophin. Many experimental therapeuticstrategies are being pursued. However, when some of these treatments transitioned into the clinical trial phase, crucial elements for their assessment were absent, including comprehensive data for natural history, meaningful outcome measures assessing clinical benefit, correlation of outcome measures with natural history, and pharmacodynamic biomarkers. During a meeting in 2009 organised by TREAT-NMD and hosted by the European Medicines Agency (EMA),³ major bottlenecks in development of treatments were identified. This meeting gave rise to a cooperative effort among stakeholders—patients and advocacy groups, academics, health-care professionals, and industry— aimed at collecting the missing data and developing thetools needed. At the same time, European Union (EU) regulators began working on guidelines to support the development of medicinal products for the treatment of Duchenne and Becker muscular dystrophies.⁴

In this Policy View, we will use the example of Duchenne muscular dystrophy to outline how the collaborative effort of stakeholders in Europe can stimulate and assist orphan medicine development in the EU, with a focus on developing functional outcome measures, biomarkers, and regulatory guidelines, and on collecting longitudinal data for natural history (panel 1). We will also discuss future aspects of treatment development for Duchenne muscular dystrophy.

Duchenne muscular dystrophy and therapeuticstrategies

Duchenne muscular dystrophy is caused by mutations in the dystrophin-encoding *DMD* gene.⁵ Without functional dystrophin, muscle fibres are susceptible to damage, resulting in chronic degeneration and replacement by connective and fat tissue, causing progressive muscle wasting and weakness.⁵⁻⁷

Up to now, one compound, ataluren, has received conditional marketing authorisation in the EU for the treatment of ambulant patients with Duchenne muscular dystrophy aged 5 years and older with a nonsense mutation (causative mutation in roughly 13% of patients).^{6,8,9} Other treatments are in clinical development, many of which have obtained orphan medicine designation in the EU (appendix). Exon skipping—the most advanced approach—aims tocorrect the disrupted reading frame in dystrophin transcripts, enabling production of a partly functional protein, as found in patients with the less progressive Becker muscular dystrophy.¹⁰ Exon skipping is induced by short, chemically modified, DNA analogues (anti- sense oligonucleotides). Because mutations cluster, skipping particular exons applies to fairly large groups of patients.^{6,9} A marketing authorisation application has been filed with the EMA for antisense oligonucleotides targeting exon 51 (applicable to about 13% of patients).^{6,9}

Care standards for Duchenne musculardystrophy

Irreversible loss of the ability to walk, self-feed, sit independently, and breathe without assisted ventilation (figure). These events are life-changing for affected children and their parents, with patients relying on full-time help in the later stages of the disease. Cardiac problems progress inevitably, leading to severe cardiomyopathy and premature death.

Standards of care have been generated and dissemi- nated in a collaborative effort of patients' organisations and TREAT-NMD, which was coordinated and supported by the US Centers for Disease Control and Prevention (CDC).¹¹⁻¹³ Multidisciplinary care (panel 2) focusing on all aspects of the disease has resulted in a slower disease progression, extending patients' mean life span to the third and fourth decade, when death generally occurs because of respiratory or heart failure. Nevertheless, in several European countries, adults and children with Duchenne muscular dystrophy receive suboptimum care,^{18,19} and care standards for adults need to be developed further.²⁰ Findings of large multicentre trials show that variability in care generates noise in outcomevariables.²¹⁻²³

Duchenne muscular dystrophy has evolved from a paediatric disease to a severe and chronic adult condition. With increasing age of patients, the management of swallowing and feeding difficulties, smooth muscle involvement in bladder and intestinal dysfunction, and issues of social integration and quality of life will need further attention. A coordinated multidisciplinary approach addressing all factors that will determine health and quality of life should be guaranteed in the transitionto adult care.

The regulatory process in the EU

Risk-benefit assessment

EU legislation requires that marketing authorisation for medicinal product is refused if the risk-benefit balanceis not deemed favourable, if therapeutic efficacy is insufficiently substantiated, or if the qualitative and quantitative composition of the medicinal product is notcontrolled appropriately. Assessment of quantified and well understood risks and benefits of a potentialtreatment is, therefore, key in the process of medicine regulation. To enable regulators to conclude on risk-benefit ratio, reliable measurements to identify and quantify risks and benefits need to be provided. Subjective judgment, input from stakeholders, andprevious decisions for other products in the same area ofmedicine also contribute to risk-benefit assessment. To make decisions as explicit and transparent as possible, regulators have adopted a systematic and structured approach to risk-benefit assessment. For products that receive marketing authorisation, the EMA provides relevant information on the risk-benefit assessment in the European Public Assessment Report (EPAR). Moreover, patients' participation in risk-benefit assessment is ensured through a framework allowing them to take an active part in regulatory workshops, scientific advisory groups, scientific advice meetings, and committee discussions.^{24,25}

Regulatory methods

Regulatory methods are in place worldwide to facilitate the development of medicinal products. Disease-specificguidelines describe regulators' preferences and standards for the demonstration of quality, safety, and efficacy of medicines. For

Duchenne muscular dystrophy, a draft guideline was published by the EMA in March, 2013, discussed among stakeholders,²⁶ and has now been published.⁴ Furthermore, the US Food and Drug Administration (FDA) has published draft guidelines.²⁷ Regulatory agencies also provide scientific advice at anystage of the development of a medicinal product to helpinvestigators conduct appropriate studies to support a future marketing authorisation. Moreover, the EU offersa range of incentives to specifically encourage the development of orphan medicines (panel 3).

For rare diseases such as Duchenne muscular dystrophy, increased uncertainty about risks and benefits is more likely to be identified at the time of the assessment. However, specific approval mechanisms exist in the EU to enable early access to medicines fulfilling an unmet medical need in a fatal disease such as Duchenne muscular dystrophy, subject to the provision of post-marketing data (eg, conditional approval).³⁰ Furthermore, the EU regulation on orphan medicinal products provides market exclusivity for 10 years for a product that has obtained a marketing authorisation.²⁸

The EMA has developed a scheme for priority medicines (PRIME) to optimise the development and accelerated assessment of medicinal products of major public health interest, such as those for rare diseases. PRIME is based on enhanced interaction and early dialogue with medicine developers; the scheme was launched in March, 2016.³¹

Outcome measures

The primary pathophysiological effect of Duchenne muscular dystrophy is a decline in muscle strength and motor function; therefore, these are important outcomes to measure. Any potential outcome measure to be used in Duchenne muscular dystrophy should be able to reliably detect and quantify a clinically meaningful effecton patients.²⁶

Functional outcome measures

Regulatory requirements in the EU postulate that an observed treatment effect needs to lead to a clear clinical benefit. Therefore, functional improvement or a delay inprogression and deterioration are judged relevantoutcomes for patients with Duchenne muscular dystrophy. To assess gross motor function, the 6-min walk test(6MWT) and the North Star Ambulatory Assessment(NSAA) are used as primary endpoints in most trials inambulant boys with Duchenne muscular dystrophy.^{32,33}

A subset of ambulant patients with behavioural andcognitive problems cannot comply with these assessments, but well defined inclusion and exclusioncriteria will help to enrol those individuals who arewilling and able to comply with all clinical trial protocolrequirements and procedures. When the first trials forDuchenne muscular dystrophy were initiated, detailedlongitudinal data for the 6MWT were scarce. Because ofcoordinated efforts of stakeholders, data are nowavailable describing the evolution in this outcome over12, 24, and 36 months in natural history studies done inItaly and Belgium and by the Cooperative InternationalNeuromuscular Research Group (CINRG).^{9,34-37} Based onthese data, we can depict longitudinal performance:young boys show some improvement in their 6MWT and NSAA scores up to the age of 7 years, butdeterioration usually takes place after this age.^{32,34,37} Similar results have been recorded with cutoff values atbaseline for the 6MWT (above or below 350 m).^{32,24,37} The combination of age and baseline cutoff values for the 6MWT has enabled identification of distincttrajectories of progression in different subgroupssubdivided by age and baseline values, which can beuseful for interpretation of clinical trial results. However, the acceptability of historical controls in clinical trials is

still a matter of discussion with regulators.^{26,38}

The rate of functional decline and its predictive value on subsequent loss of ambulation has been established for the NSAA from a large database (UK North Star network) and for the 6MWT using data from CINRG, and this work was useful for postulation of the expected minimal clinically important difference (MCID).^{32,34,37,39} Furthermore, knowing the rate of decline and expected variation enables stratification and power calculations. Any target effect size should be discussed in advance with regulators to define expectations and agree on what constitutes a clinically relevant change in a given experimental setting. Evaluation of quality of life is an important aspect of treatment assessment. In patients with Duchenne muscular dystrophy, a strong correlation was noted between the 6MWT and the global Pediatric Outcome Data Collection Instrument (PODCI)—a health-related quality-of-life measure of functional ability. Even at highlevels of disability, small increases in the 6MWT lead to meaningful changes in quality-of-life scores.³⁹

New functional outcome measure scales

By definition, the 6MWT and NSAA cannot be used in non-ambulant individuals. Since the average age at loss of ambulation is roughly 10.5 years,⁴⁰ and median survival of patients is about 30 years,^{41,42} most people with Duchenne muscular dystrophy are non-ambulant. Toaddress this issue, a collaborative international group—including boys with Duchenne muscular dystrophy and their families—developed the Performance of Upper Limb scale to assess upper limb function in ambulant and non-ambulant patients with the disease.^{43,45} The scale has been validated for clinical use against other functional measures (eg, the 6MWT),⁴⁶ and longitudinal data are emerging for ambulant and non-ambulant patients, withand without use of steroids.⁴⁷ The scale is awaiting regulatory acceptance.

Studies have been done with neurodevelopmental scales in young boys with Duchenne muscular dystrophy, evenfrom the neonatal period.⁴⁸⁻⁵⁰ Findings show that these boys have delayed motor milestones, most strikingly in thegross locomotor and language subscales, and that the gap with age-matched peers increases with age for motor skills. This result has led to the understanding among stakeholders that, should therapeutic interventions be proven effective and safe, it would be important to administer them as early as possible.

Biomarkers and surrogate endpoints

Biomarkers are important to inform and guide medicinedevelopment, and they have regulatory applications—eg, to confirm the mechanism of action (pharmacodynamicsbiomarkers). When a clear relation with clinical outcomes has been established, biomarkers can even be used as primary outcome measures (surrogate endpoints) instead of a functional outcome measure. Because biomarkers are measured objectively they are less prone to variation from factors such as motivation and compliance with functional tests. However, to fit with regulatory requirements, biomarkers must be validated for a specific context of use—eg, trial enrichment or surrogate endpoint. A dedicated procedure is in place at the EMA for qualification of biomarkers and novel methods to use in the context of research and the development of drugs.^{51,52}

Dystrophin

Production of dystrophin protein was deemed an obvious choice for a pharmacodynamics marker in trials of a compound aimed at re-expression of dystrophin.⁵³ Moreover, detection of dystrophin was a secondary endpoint in early-phase dose-escalation studies of exon skipping treatments and ataluren.⁵⁴⁻⁵⁸ In practice, however, quantification of dystrophin is not straight- forward.⁵³ To use dystrophin as a pharmacodynamics biomarker, methods to quantify dystrophin must be proven reliableand reproducible. Efforts of an international working group have shown that, by using a carefully devisedstandard operating procedure, and by sharing (in a blinded fashion) the same material, patients with different levels of dystrophin production can be stratified accurately, with good intralaboratory and interlaboratory reliability and with high correlation between western blot and immunocytochemistry,⁵⁹ using several dystrophin quantification protocols.⁶⁰⁻⁶² Further improvements to decrease the coefficients of variation (particularly for low dystrophin levels) for these techniques are an important next step in validating dystrophin as a pharmacodynamics biomarker for therapeutic efficacy. Up to now, data have been insufficient to establish a clear correlation between dystrophin levels and muscle function for various stages of disease. Thus, use of dystrophin as a surrogate primary endpoint is questionable.

MRI

Techniques in MRI and magnetic resonance spectroscopy are promising for quantification of disease pathology and progression in a non-invasive and longitudinal fashion. Protocols have been developed and validated on several different magnetic resonance platforms to measure muscle oedema and inflammation in Duchenne muscular dystrophy.⁶³⁻⁶⁶ Now that protocols are validated across platforms and sites, MRI and magnetic resonance spectroscopy can be used as quantitative—and in most cases exploratory—outcome measures in several ongoingnatural history studies and intervention trials. Specialisedprotocols to assess treatment effects quantitatively havebeen tested independently across neuromuscular centresworldwide, and promising results have been published.⁶³⁻⁶⁷The ImagingDMD consortium in the USA, which is supported by various patients' organisations and the USNational Institutes of Health, has collected longitudinaldata for a large cohort of patients with Duchennemuscular dystrophy. Their findings show that magnetic resonance measures of T2 and lipid fraction have goodsensitivity to detect Duchenne muscular dystrophydisease pathology and progression, even in young boys inwhom functional outcomes improve with time.⁶⁸Furthermore, MRI and magnetic resonance spectroscopycan detect the therapeutic effects of corticosteroids in reducing inflammatory processes in skeletal muscles ofboys with Duchenne muscular dystrophy.⁶⁹ As such, MRIshows promise as a surrogate outcome measure, although more data for natural history need to be collected.

Extrapolation

Because of the effect of disease stage and age on functional outcome measures for Duchenne muscular dystrophy, having well defined and homogeneous cohorts of patients in clinical trials is important. This cohort specificity can reduce patients' variability in function, which is crucial for reliable identification of a treatment effect in a specific population. However, it can also affect the indication for which the drug canpotentially be approved, because sufficient evidence needs to be available to allow for a separate risk-benefitratio conclusion in other subgroups of patients (eg, per disease stage, or ambulant *vs* non-ambulant).

Up to now, most trials in Duchenne muscular dystrophy have been done in patients who can comply with the 6MWT—ie, ambulant individuals aged 5 years and older (about 20–25% of patients). However, early treatment is anticipated to lead to a larger therapeutic effect. Nevertheless, non-ambulant patients would certainly also benefit from slower deterioration of their residual muscle function (ie, motor, respiratory, and cardiac) and, therefore, a drug indication for both ambulant and non-ambulant individuals would be a preferable goal.

Extrapolation of data from a trial done in a specific subgroup of patients to a different population should bediscussed with regulators on a case-by-case basis. The current position of the EMA is that, if supported by the mechanism of action, extrapolation of data from older toyounger (or from younger to older) patients might be discussed in the context of additional real-life data needed to be collected after authorisation. When data are generated in a subset of the population, to obtain a broadlicense for a product, further data will be needed inpatients outside this subset to address any uncertainties These aspects are discussed increasingly but, for further consideration by a committee for human medicinal products (CHMP), scientific advice should be sought to discuss the most appropriate strategy for development.

Future perspectives

Since their first meeting with European regulators in 2009,³ the academic and patient communities in Duchenne muscular dystrophy have become more aware of regulatory processes. In collaboration withpharmaceutical companies working in the area ofDuchenne muscular dystrophy, these stakeholders havetried to address gaps in knowledge identified during themeeting.⁷⁰ Large amounts of data have been collected and new outcome measures and tools have beendeveloped,

building on existing resources—eg, patient registries—provided by patients' organisations and TREAT-NMD.^{6,71-73} At the same time, regulators have become more familiar with the specifics of developing new medicines for Duchenne muscular dystrophy and have finalised guidelines on product development for Duchenne and Becker muscular dystrophies.⁴ This improved mutual understanding has been helpful for a continuous and constructive dialogue that has moved the area forward. A stakeholder meeting held in April, 2015, in London, UK, allowed for further alignment of ongoingwork and prioritisation of future efforts.

First, efforts to increase international awareness of care standards for Duchenne muscular dystrophy need tocontinue, because patients deserve access to optimum care. Plans to set up a European Reference Network for neuromuscular disorders will build on the TREAT-NMDcare and trial site registry (CTSR)⁷² and the CARE-NMD project¹⁹ and facilitate implementation of care standardsfor Duchenne muscular dystrophy throughout Europe. This work would complement efforts Parent Project Muscular Dystrophy (PPMD) is currently coordinating in the USA to certify centres that provide care according to international guidelines.⁷⁴

Second, new centres participating in trials are needed.Many clinical trials are currently done in the area of Duchenne muscular dystrophy, resulting in capacity problems at experienced trial sites. Adhering to care standards is a prerequisite to be selected as a trial site bypharmaceutical companies.

Third, another PPMD-led initiative is defining core sets of outcome measures to be used in ambulant and non- ambulant patients, which ideally should be used in all trials for Duchenne muscular dystrophy. This stand- ardisation would facilitate the trial process, because personnel will have to be trained once rather than for every trial. Furthermore, it would allow comparison of results between different trials and facilitate post- marketing surveillance.

Fourth, regulators offer scientific and regulatory guidance. Platforms are available to discuss specific medicine development, development of biomarkers,

functional outcome measures, and patient-reportedoutcomes, for example. Through increased dialogue, advice can be sought from the EMA towards qualifi-cation of outcome measures in Duchenne muscular dystrophy—eg, the Performance of Upper Limb scale as a functional outcome measure in non-ambulant patients, and MRI as a biomarker or surrogate endpoint for Duchenne muscular dystrophy. The same platform couldbe considered for the quantification of dystrophin expression as a pharmacodynamic biomarker, which hasbeen discussed at a workshop organised by the FDA and the National Institutes of Health.⁷⁵

Fifth, developing a treatment for a rare disease such asDuchenne muscular dystrophy should be a global effort.Regulatory requirements should be aligned and commun-ication should be continuous between regulatory bodies in different global regions with respect to guidelines for treatment development and biomarker qualification.⁷⁶

Sixth, publication of data in peer-reviewed journals is important, because this process informs the scientific community and regulatory bodies, allowing data to be used in guideline development, scientific advice, and medicine assessment. Data have been published for natural history and functional outcome measures in ambulant patients, and researchers have started publishing data for MRI as a potential biomarker for muscle quality. Focus should now be on producing natural history data and outcome measures for non-ambulant patients—eg, upper-limb function scales and heart and respiratory function.

Seventh, placebo-controlled trials are needed to study safety and efficacy of new treatments. However, future use of natural history data or data from the placebo armof other trials has not been excluded. Several large, natural history studies are underway (eg, NCT01753804, NCT01385917, and NCT00468832): it is important to align the outcome measures used in ongoing natural history studies and clinical trials and for the research groups involved to share their datasets. Several initiatives to collect and curate datasets are ongoing.

Finally, most clinical trials are done in selected populations of patients with Duchenne muscular dystrophy, generally in ambulant patients. However, to allow extrapolation of efficacy and safety data to obtain abroader indication (eg, for all patients with Duchenne muscular dystrophy rather than a focus on a specific group of ambulant boys), collection of data to validate the extrapolation is crucial. Data obtained in patients outside the inclusion criteria of the trial population are needed; collection of natural history data and development of outcome measures will be increasingly important, as wellas for effective assessments in post-marketing studies.

Concluding remarks

The collaborative effort of researchers, health-care professionals, representatives from industry, regulators, and patients has been instrumental in moving the area of Duchenne muscular dystrophy forward in Europe (panel 4). In parallel, comparable efforts are ongoing in the USA (eg, the Action Plan for Muscular Dystrophies),^{77,78} and the FDA has programmes for clinical outcome assessment, biomarker qualification, and regulatory guidance.⁷⁹⁻⁸¹ Nevertheless, the work is not yet complete and eight new focus areas have been identified. Each of these priority areas will need continued involvement from stakeholders. Although these tasks might seem challenging, a strong basis of previous work, mutual understanding, and collabor- ation will aid these efforts. Previous work focused mainly on doing trials to obtain marketing authorisation; however, research has now started to address challenges around post-marketing and treatment access strategies.



References

- 1 Mendell JR, Shilling C, Leslie ND, et al. Evidence-based path to newborn screening for Duchenne muscular dystrophy. Ann Neurol 2012; 71: 304–13.
- 2 Moat SJ, Bradley DM, Salmon R, Clarke A, Hartley L. Newborn bloodspot screening for Duchenne muscular dystrophy:
- 3 Muntoni F. The development of antisense oligonucleotide therapies for Duchenne muscular dystrophy: report on a TREAT-NMD workshop hosted by the European Medicines Agency (EMA), on September 25th 2009. *Neuromuscul Disord* 2010; **20**: 355–62.
- 4 European Medicines Agency, Committee for Medicinal Products for Human Use. Guideline on the clinical investigation of medicinal products for the treatment of Duchenne and Becker muscular dystrophy (EMA/CHMP/236981/2011, Corr. 1). Dec 17, 2015. http://www.ema.europa.eu/docs/en_GB/document_library/Scientific_guideline/2015/12/WC500199239.pdf (accessed April 12, 2016).
- 5 Hoffman EP, Brown RH, Kunkel LM. Dystrophin: the proteinproduct of the Duchenne muscular dystrophy locus—1987. Biotechnology 1992; 24: 457-66.
- 6 Bladen CL, Salgado D, Monges S, et al. The TREAT-NMD DMD Global Database: analysis of more than 7,000 Duchenne musculardystrophy mutations. Hum Mutat 2015; 36: 395–402.
- 7 Grounds MD. Two-tiered hypotheses for Duchenne musculardystrophy. Cell Mol Life Sci 2008; 65: 1621–25.
- 8 Guiraud S, Aartsma-Rus A, Vieira NM, Davies KE, van Ommen GJ, Kunkel LM. The pathogenesis and therapy of muscular dystrophies. *Annu Rev Genomics Hum Genet* 2015; **16**: 281–308.
- 9 Aartsma-Rus A, Fokkema I, Verschuuren J, et al. Theoretic applicability of antisense-mediated exon skipping for Duchenne muscular dystrophy mutations. Hum Mutat 2009; 30: 293–99.
- 10 Aartsma-Rus A. Antisense-mediated modulation of splicing: therapeutic implications for Duchenne muscular dystrophy. RNA Biol 2010; 7: 453–61.
- 11 Bushby K, Finkel R, Birnkrant DJ, et al, for the DMD Care Considerations Working Group. Diagnosis and management of Duchenne muscular dystrophy, part 2: implementation of multidisciplinary care. *Lancet Neurol* 2010; **9**: 177–89.
- 12 Bushby K, Finkel R, Birnkrant DJ, et al, for the DMD Care Considerations Working Group. Diagnosis and management of Duchenne muscular dystrophy, part 1: diagnosis, and pharmacological and psychosocial management. *Lancet Neurol*2010; **9**: 77–93.
- 13 Kinnett K, Rodger S, Vroom E, Furlong P, Aartsma-Rus A, Bushby K. Imperatives for DUCHENNE MD: a simplified guide to comprehensive care for Duchenne muscular dystrophy. *PLoS Curr* 2015; published online Aug 7. DOI:10.1371/currents.md.87770501e86 f36f1c71e0a5882ed9ba1.
- 14 TREAT-NMD Neuromuscular Network. Family guide in different languages. Feb 3, 2016. http://www.treat-nmd.eu/care/dmd/family-guide/translations (accessed April 15, 2016).
- 15 TREAT-NMD Neuromuscular Network. The imperatives of DMD. Sept 11, 2015. http://www.treat-nmd.eu/care/dmd/imperatives-dmd(accessed April 15, 2016).
- 16 Duboc D, Meune C, Lerebours G, Devaux JY, Vaksmann G, Bécane HM. Effect of perindopril on the onset and progression ofleft ventricular dysfunction in Duchenne muscular dystrophy. J Am Coll Cardiol 2005; 45: 855–57.
- 17 Duboc D, Meune C, Pierre B, et al. Perindopril preventive treatmenton mortality in Duchenne muscular dystrophy: 10 years' follow-up.*Am Heart J* 2007; **154**: 596–602.
- 18 Landfeldt E, Lindgren P, Bell CF, et al. Compliance to care guidelines for Duchenne muscular dystrophy. J Neuromusc Dis2015; 2: 63–72.
- 19 Rodger S, Woods KL, Bladen CL, et al. Adult care for Duchennemuscular dystrophy in the UK. J Neurol 2015; 262: 629–41.
- 20 Rahbek J, Steffensen BF, Bushby K, de Groot IJ. 206th ENMC International Workshop: care for a novel group of patients—adultswith Duchenne muscular dystrophy, Naarden, The Netherlands, 23–25 May 2014. *Neuromuscul Disord* 2015; **25**: 727–38.
- 21 Ricotti V, Ridout DA, Scott E, et al. Long-term benefits and adverseeffects of intermittent versus daily glucocorticoids in boys with Duchenne muscular dystrophy. J Neurol Neurosurg Psychiatry 2013;84: 698–705.
- 22 Griggs RC, Herr BE, Reha A, et al. Corticosteroids in Duchenne muscular dystrophy: major variations in practice. Muscle Nerve 2013;48: 27–31.
- Strehle EM, Straub V. Recent advances in the management of Duchenne muscular dystrophy. *Arch Dis Child* 2015; 100: 1173–77.
 European Medicines Agency. Patients and consumers. http://www.ema.europa.eu/ema/index.jsp?curl=pages/partners_and_networks/
- general/general_content_000317.jsp&mid=WC0b01ac058003500c (accessed April 13, 2016).
 European Medicines Agency. Scientific advice and protocol assistance. http://www.ema.europa.eu/ema/index.isp?curl=pages/
- 25 European Medicines Agency. Scientific advice and protocol assistance. http://www.ema.europa.eu/ema/index.jsp?curl=pages/ regulation/general/general_content_000049. jsp&mid=WC0b01ac05800229b9 (accessed April 13, 2016).
- 26 Lynn S, Aartsma-Rus A, Bushby K, et al. Measuring clinical effectiveness of medicinal products for the treatment of Duchennemuscular dystrophy.

Neuromuscul Disord 2015; 25: 96–105.

- 27 Department of Health and Human Services, Food and Drug Administration, Center for Drug Evaluation and Research (CDER).Duchenne muscular dystrophy and related dystrophinopathies: developing drugs for treatment—guidance for industry. May 22, 2015. http://www.fda.gov/downloads/Drugs/ GuidanceComplianceRegulatoryInformation/Guidances/ UCM450229.pdf (accessed April 25, 2016).
- European Parliament, Council of the European Union.Regulation (EC) no 141/2000 of the European Parliament and of theCouncil of 16 December 1999 on orphan medicinal products. http://ec.europa.eu/health/files/eudralex/vol-1/reg_2000_141/ reg_2000_141_en.pdf (accessed April 13, 2016).
- 29 European Medicines Agency. Orphan designation. http://www.ema.europa.eu/ema/index.jsp?curl=pages/regulation/general_content_000029.jsp&mid=WC0b01ac05800240ce (accessed April 15, 2016).
- Aartsma-Rus A, Ferlini A, Goemans N, et al. Translational and regulatory challenges for exon skipping therapies. *Hum Gene Ther*2014; **25:** 885–92. European Medicines Agency. PRIME: priority medicines. 2016. http://www.ema.europa.eu/ema/index.jsp?curl=pages/regulation/
- general/general_content_000660.jsp&mid=WC0b01ac05809f8439 (accessed April 25, 2016).
- 32 Ricotti V, Ridout DA, Pane M, et al, on behalf of UK NorthStar Clinical Network. The NorthStar Ambulatory Assessment in Duchenne muscular dystrophy: considerations for the design of clinical trials. J Neurol Neurosurg Psychiatry 2016; 87: 149–55.
- 33 McDonald CM, Henricson EK, Han JJ, et al. The 6-minute walk testas a new outcome measure in Duchenne muscular dystrophy. *Muscle Nerve* 2010; **41**: 500–10.
- 34 Mazzone ES, Pane M, Sormani MP, et al. 24 month longitudinaldata in ambulant boys with Duchenne muscular dystrophy. *PLoS One* 2013; 8: e52512.
- 35 Pane M, Mazzone ES, Sivo S, et al. Long term natural history datain ambulant boys with Duchenne muscular dystrophy: 36-month changes. *PLoS* One 2014; 9: e108205.
- 36 Pane M, Mazzone ES, Sormani MP, et al. 6 minute walk test inDuchenne MD patients with different mutations: 12 month changes. *PLoS One* 2014; 9: e83400.
- 37 McDonald CM, Henricson EK, Abresch RT, et al. The 6-minute walk test and other endpoints in Duchenne muscular dystrophy:longitudinal natural history observations over 48 weeks from a multicenter study. *Muscle Nerve* 2013; **48**: 343–56.
- 38 Mercuri E, Mayhew A, Muntoni F, et al. Towards harmonisation of outcome measures for DMD and SMA within TREAT-NMD, report of three expert workshops: TREAT-NMD/ENMC workshop on outcome measures, 12th–13th May 2007, Naarden, The Netherlands; TREAT-NMD workshop on outcome measures in experimental trials for DMD, 30th June–1st July 2007, Naarden, The Netherlands; conjoint Institute of Myology TREAT-NMD meeting on physical activity monitoring in neuromuscular disorders, 11th July 2007, Paris, France. *Neuromuscul Disord* 2008; 18: 894–903.
- 39 Henricson EK, Abresch RT, Cnaan A, et al. The Cooperative International Neuromuscular Research Group Duchenne natural history study: glucocorticoid treatment preserves clinically meaningful functional milestones and reduces rate of disease progression as measured by manual muscle testing and other commonly used clinical trial outcome measures. *Muscle Nerve* 2013; **48**: 55–67.
- 40 Wang RT, Silverstein Fadlon CA, Ulm JW, et al. Online self-report data for Duchenne muscular dystrophy confirms natural history and can be used to assess for therapeutic benefits. *PLoS Curr* 2014; published online Oct 17. DOI:10.1371/currents.md. e1e8f2be7c949f9ffe81ec6fca1cce6a.
- Passamano L, Taglia A, Palladino A, et al. Improvement of survivalin Duchenne muscular dystrophy: retrospective
- analysis of 835 patients. Acta Myol 2012; 31: 121–25.
- 42 Kohler M, Clarenbach CF, Bahler C, Brack T, Russi EW, Bloch KE.Disability and survival in Duchenne muscular dystrophy.

J Neurol Neurosurg Psychiatry 2009; 80: 320-25.

- 43 Mayhew A, Mazzone ES, Eagle M, et al. Development of the performance of the upper limb module for Duchenne musculardystrophy. *Dev Med Child Neurol* 2013; **55**: 1038–45.
- 44 Mercuri E, McDonald C, Mayhew A, et al. International workshopon assessment of upper limb function in Duchenne muscular dystrophy: Rome, 15–16 February 2012. *Neuromuscul Disord* 2012; **22**: 1025–28.
- 45 Pane M, Mazzone ES, Fanelli L, et al. Reliability of the performance of upper limb assessment in Duchenne muscular dystrophy. *Neuromuscul Disord* 2014; 24: 201–06.
- 46 Pane M, Mazzone ES, Sivo S, et al. The 6 minute walk test and performance of upper limb in ambulant Duchenne muscular dystrophy boys. *PLoS Curr* 2014; published online Oct 7. DOI:10.1371/currents.md.a93d9904d57dcb08936f2ea89bca6fe6.
- 47 Pane M, Fanelli L, Mazzone ES, et al. Benefits of glucocorticoids innon-ambulant boys/men with Duchenne muscular dystrophy:

a multicentric longitudinal study using the performance of upperlimb test. *Neuromuscul Disord* 2015; 25: 749–53.

- 48 Connolly AM, Florence JM, Cradock MM, et al. Motor and cognitive assessment of infants and young boys with Duchenne muscular dystrophy: results from the Muscular Dystrophy Association DMD Clinical Research Network. *Neuromuscul Disord* 2013; 23: 529–39.
- 49 Connolly AM, Florence JM, Cradock MM, et al. One year outcomeof boys with Duchenne muscular dystrophy using the Bayley-III scales of infant and toddler development. *Pediatr Neurol* 2014; 50: 557–63.
- 50 Pane M, Scalise R, Berardinelli A, et al. Early neurodevelopmental assessment in Duchenne muscular dystrophy. Neuromuscul Disord2013; 23: 451-55.
- 51 Ferlini A, Flanigan KM, Lochmuller H, Muntoni F, 't Hoen PAC, McNally E. 204th ENMC International Workshop on Biomarkers inDuchenne Muscular Dystrophy 24-26 January 2014, Naarden, The Netherlands. *Neuromuscul Disord* 2015; 25: 184–98.
- 52 Aartsma-Rus A, Ferlini A, Vroom E. Biomarkers and surrogate endpoints in Duchenne: meeting report. *Neuromuscul Disord* 2014; 24: 743–45.
- 53 Aartsma-Rus A. Dystrophin analysis in clinical trials. *J Neuromusc Dis* 2014; **1**: 41–53.
- 54 Cirak S, Arechavala-Gomeza V, Guglieri M, et al. Exon skipping anddystrophin restoration in patients with Duchenne muscular dystrophy after systemic phosphorodiamidate morpholino oligomer treatment: an open-label, phase 2, doseescalation study. *Lancet* 2011; 378: 595–605.
- 55 Goemans NM, Tulinius M, van den Akker JT, et al. Systemic administration of PRO051 in Duchenne's musculardystrophy. N Engl J Med 2011; 364: 1513–22.
- 56 Kinali M, Arechavala-Gomeza V, Feng L, et al. Local restoration ofdystrophin expression with the morpholino oligomer AVI-4658 in Duchenne muscular dystrophy: a single-blind, placebo-controlled, dose-escalation, proof-of-concept study. *Lancet Neurol* 2009; 8: 918–28.
- 57 Van Deutekom JC, Janson AA, Ginjaar IB, et al. Local dystrophin restoration with antisense oligonucleotide PRO051. *N Engl J Med*2007; **357:** 2677–86.
- 58 Finkel RS, Flanigan KM, Wong B, et al. Phase 2a study of ataluren-mediated dystrophin production in patients with nonsense mutation Duchenne muscular dystrophy. *PLoS One*2013; **8**: e81302.

- 59 Anthony K, Arechavala-Gomeza V, Taylor LE, et al. Dystrophin quantification: biological and translational research implications. *Neurology* 2014; 83: 2062–69.
- 60 Arechavala-Gomeza V, Kinali M, Feng L, et al. Immunohistologicalintensity measurements as a tool to assess sarcolemmaassociated protein expression. *Neuropathol Appl Neurobiol* 2010; 36: 265–74.
- 61 Taylor LE, Kaminoh YJ, Rodesch CK, Flanigan KM. Quantification of dystrophin immunofluorescence in dystrophinopathy muscle specimens. *Neuropathol Appl Neurobiol* 2012; 38: 591–601.
- Beekman C, Sipkens JA, Testerink J, et al. A sensitive, reproducibleand objective immunofluorescence analysis method of dystrophinin individual fibers in samples from patients with Duchenne muscular dystrophy. *PLoS One* 2014; 9: e107494.
 Finanger EL, Russman B, Forbes SC, Rooney WD, Walter GA, Vandenborne K. Use of skeletal muscle MRI in diagnosis
- and monitoring disease progression in Duchenne muscular dystrophy. *Phys Med Rehabil Clin N Am* 2012; 23: 1–10.
 Fischmann A, Hafner P, Gloor M, et al. Quantitative MRI and loss of free ambulation in Duchenne muscular dystrophy. *J*
- Fischmann A, Hamer P, Gloor M, et al. Quantitative MRI and loss of free ambulation in Duchenne muscular dystrophy. J Neurol 2013;260: 969–74.
 Hollingsworth KG, Garrood P, Eagle M, Bushby K, Straub V. Magnetic resonance imaging in Duchenne muscular
- Holingsworth KG, Garrood P, Eagle M, Bushby K, Straub V. Magnetic resonance imaging in Duchenne muscular dystrophy:longitudinal assessment of natural history over 18 months. *Muscle Nerve* 2013; 48: 586–88.
- 66 Willcocks RJ, Arpan IA, Forbes SC, et al. Longitudinal measurements of MRI-T2 in boys with Duchennemuscular dystrophy: effects of age and disease progression. *Neuromuscul Disord* 2014; 24: 393–401.
- 67 Forbes SC, Lott DJ, Finkel RS, et al. MRI/MRS evaluation of afemale carrier of Duchenne muscular dystrophy. Neuromuscul Disord 2012; 22 (suppl 2): S111–21.
- 68 Willcocks RJ, Rooney WD, Triplett WT, et al. Multicenter prospective longitudinal study of magnetic resonancebiomarkers in a large duchenne muscular dystrophy cohort. Ann Neurol 2016; 79: 535–47.
- 69 Arpan I, Willcocks RJ, Forbes SC, et al. Examination of effects of corticosteroids on skeletal muscles of boys with DMD using MRIand MRS. *Neurology* 2014; 83: 974–80.
- 70 European Cooperation in Science and Technology (COST), Exon Skipping in Duchenne Muscular Dystrophy (SCOPE-DMD). Briefing document: antisense oligonucleotide-mediated exon skipping therapy development for Duchenne muscular dystrophy (DMD)—a COST/SCOPE-DMD meeting hosted by the European Medicines Agency. April, 2015. http://exonskipping.eu/wp-content/uploads/2015/04/Briefing-Document-COST-and-SCOPE-DMD- EMA-meeting-April-2015-FINAL-VERSION.pdf (accessed
- 71 Sejerson T, Bushby K. Standards of care for Duchenne muscular dystrophy: brief TREAT-NMD recommendations. *Adv Med Biol*2009; **652:** 13–21.